Author's response to reviews

Title: A qualitative study of Chinese Canadian fathers' smoking behaviors: Intersecting cultures and masculinities

Authors:

Aimei Mao (maoaimei@kwnc.edu.mo)
Joan L. Bottorff (joan.bottorff@ubc.ca)
John L. Oliffe (john.oliffe@nursing.ubc.ca)
Gayle Sarbit (gayl.sarbit@ubc.ca)
Mary T Kelly (mary.theresa.kelly@gmail.com)

Version: 4 Date: 5 March 2015

Author's response to reviews: see over
Dear Dr. Silvestre

MS: 2087337509149519
A qualitative study of Chinese Canadian fathers' smoking behaviors: Intersecting cultures and masculinities Aimei Mao, Joan L. Bottorff, John L. Oliffe, Gayle Sarbit and Mary T Kelly

Thank you for your review of this manuscript. Please have made the following revisions (highlighted in yellow in the revised manuscript):

<table>
<thead>
<tr>
<th>Requested changes</th>
<th>Revisions made</th>
</tr>
</thead>
<tbody>
<tr>
<td>In introduction (p 5, line 117), please remove &quot;and limited awareness of its health consequences&quot;, unless you have a scientific basis for this statement (then please add this).</td>
<td>We have added a reference to support this statement.</td>
</tr>
<tr>
<td>There's a small spelling error in your ethical board statement, please revise.</td>
<td>We have corrected the spelling error in “university”</td>
</tr>
<tr>
<td>Please remove the recommendations with regard to studies in China, as this does not necessarily follow from your data (your study sample has not lived in China for at least six months). Focus on how this information can be used to influence your study population instead.</td>
<td>We have removed the recommendations as requested and revised this para to focus on the study population.</td>
</tr>
<tr>
<td>Please also ensure that your revised manuscript conforms to the journal style (<a href="http://www.biomedcentral.com/info/ifora/medicine_journals">http://www.biomedcentral.com/info/ifora/medicine_journals</a>).</td>
<td>Revisions have been made in formatting to conform to the journal style.</td>
</tr>
</tbody>
</table>

Please let us know if you need any additional information.

Sincerely

Aimei Mao
Joan Bottorff